Author's response to reviews

Title: Canine retraction and anchorage loss self-ligating versus conventional brackets: A systematic review and meta-analysis

Authors:

Qiaozhen Zhou Ms (772014326@qq.com)
Kui Huang Dr (382829810@qq.com)
Yu Zhou Dr (wzmczy@126.com)
Liu Tian Dr (luobo337@126.com)
Xiaofeng Chen Ms (cxf120120@163.com)
Abdul A Haq Dr (azeem12691@gmail.com)

Version: 4 Date: 29 September 2015

Author's response to reviews:

Author's response to reviews
Title: Canine retraction and anchorage loss self-ligating versus conventional brackets: A systematic review and meta-analysis
Authors#
Qiaozhen Zhou #772014326@qq.com#
Abdul Azeem Amin ul Haq (azeem12691@gmail.com)
Liu Tian #185137393@qq.com#
Xiaofeng Chen #cxf120120@163.com#
Kui Huang #7933471@qq.com#
Yu Zhou #wzmczy@126.com#

Version: 2 Date: 30 September 2015

Author's response to reviews: see over

1. Additional authors
We note that since the original submission, three new names have been added to the authors list (Qiaozhen Zhou, Abdul Azeem Amin ul Haq and Liu Tian). We can therefore not proceed until all authors, including the newly added ones, send us an email confirming that they are happy with this change of the list of authors.
Respond: I have let they send you an email confirming that they are happy with this change of the list of authors.

2. Email addresses
Please provide the email addresses of all authors on the title page.
Respond: we added it.
3. Abstract
Please change subsection title 'Objective' of the abstract to 'Background' and add a description of the context of your study. Please also change subsection 'Materials and Methods' to only 'Methods'. Please do these changes in both the manuscript and the abstract that is separately uploaded in the submission system.

Respond: we revise it according to your advices

4. Abbreviations
Please ensure that all abbreviations used are defined in the text at first use. Currently, TAD for example is undefined. Please also add a section "List of abbreviations used", which should precede "Competing interests" and "Authors' contributions".

Respond: we added it.

5. Acknowledgements
Please add a section "Acknowledgements" before the reference list. There you can acknowledge anyone who made substantial contributions towards the article, but who does not meet the criteria for authorship, and the source(s) of funding for each author.

Respond: we added it.

6. Tables
Please check whether it is indeed table 2 that should be referred to in subsection 'Risk of bias', and not table 3.

Respond: I am sorry for this mistake. It is table 3 not table 2.